Author’s response to reviews

Title: Guidelines disobeyed - A severe complication associated with manual therapy of the cervical spine in a patient with Forestier's disease: a case report

Authors:

Maximilian J Hartel (mh@maxhartel.de)
Ulrich Seidel (ulrich.seidel@insel.ch)
Lukas Iselin (iselinlukas@mac.com)
Aristomenis K Exadaktylos (exadaktylos@exadaktylos.ch)
Lorin M Benneker (lorinmichael.benneker@insel.ch)

Version: 4 Date: 12 September 2011

Author’s response to reviews: see over
Dear Reviewers,

We would like to thank you again for the constructive review that led to a significant improvement of the paper. Please find in the following the answer to your requests.

Sincerely yours

[Signature]

Maximilian J. Hartel
Corresponding Author
Answers to reviewer’s comments:

Reviewer: Vijayakumar Javalkar

Comments to authors:
Though the authors have addressed the reference for roger kerry statement it has not been incorporated in the revised manuscript.
Please quote the reference for roger kerry statement in the manuscript and update the references. The submitted manuscript doesn't reflect this changes.

Answer:
Mr. Javalkar is right – the reference was missing. Roger Kerry’s statement has now been referenced. Please see page 5, line 25.
Reviewer: Thomas H Wuerz

Comments to authors:
The revision of the manuscript resulted in a significant improvement.

Answer: Thank you for the extensive feedback that helped to improve the report significantly!
Reviewer: Haymo G Thiel

Comments to authors:
I am satisfied with the revisions undertaken by the authors and the answers provided to my suggestions

Answer: Thank you for the detailed feedback that helped to improve the report significantly!